

Citation: Tiffany A, Dalziel BD, Kagume Njenge H, Johnson G, Nugba Ballah R, James D, et al. (2017) Estimating the number of secondary Ebola cases resulting from an unsafe burial and risk factors for transmission during the West Africa Ebola epidemic. PLoS Negl Trop Dis 11(6): e0005491. https://doi.org/10.1371/journal.pntd.0005491

Editor: Oladele B. Akogun, Common Heritage Foundation, NIGERIA

Received: January 4, 2017

Accepted: March 14, 2017

Published: June 22, 2017

Copyright: © 2017 Tiffany et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper.

Funding: This study was funded by the International Federation of the Red Cross and Red Crescent Societies. AM participated in study design and manuscript preparation, HKN and AW participated in data collection and manuscript preparation. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. RESEARCH ARTICLE

Estimating the number of secondary Ebola cases resulting from an unsafe burial and risk factors for transmission during the West Africa Ebola epidemic

Amanda Tiffany¹*, Benjamin D. Dalziel^{2,3}, Hilary Kagume Njenge⁴, Ginger Johnson⁵, Roselyn Nugba Ballah⁶, Daniel James⁷, Abdoulaye Wone⁸, Juliet Bedford⁹, Amanda McClelland¹⁰

 Epicentre, Geneva, Switzerland, 2 Department of Integrative Biology, Oregon State University, Corvallis, Oregon, United States of America, 3 Department of Mathematics, Oregon State University, Corvallis, Oregon, United States of America, 4 International Federation of the Red Cross and Red Crescent Societies, Monrovia, Liberia, 5 Anthrologica, Dallas, Texas, United States of America, 6 Liberian Red Cross, Monrovia, Liberia, 7 Sierra Leone Red Cross Society, Freetown, Sierra Leone, 8 International Federation of the Red Cross and Red Crescent Societies, Conakry, Guinea, 9 Anthrologica, Oxford, United Kingdom,
International Federation of the Red Cross and Red Crescent Societies, Geneva, Switzerland

* amanda.tiffany@geneva.msf.org

Abstract

Background

Safely burying Ebola infected individuals is acknowledged to be important for controlling Ebola epidemics and was a major component of the 2013–2016 West Africa Ebola response. Yet, in order to understand the impact of safe burial programs it is necessary to elucidate the role of unsafe burials in sustaining chains of Ebola transmission and how the risk posed by activities surrounding unsafe burials, including care provided at home prior to death, vary with human behavior and geography.

Methodology/Principal findings

Interviews with next of kin and community members were carried out for unsafe burials in Sierra Leone, Liberia and Guinea, in six districts where the Red Cross was responsible for safe and dignified burials (SDB). Districts were randomly selected from a district-specific sampling frame comprised of villages and neighborhoods that had experienced cases of Ebola. An average of 2.58 secondary cases were potentially generated per unsafe burial and varied by district (range: 0–20). Contact before and after death was reported for 142 (46%) contacts. Caregivers of a primary case were 2.63 to 5.92 times more likely to become EVD infected compared to those with post-mortem contact only. Using these estimates, the Red Cross SDB program potentially averted between 1,411 and 10,452 secondary EVD cases, reducing the epidemic by 4.9% to 36.5%.

Conclusions/Significance

SDB is a fundamental control measure that limits community transmission of Ebola; however, for those individuals having contact before and after death, it was impossible to



Competing interests: I have read the journal's policy and the authors of the manuscript have the following competing interests: Epicentre (AT) and Anthrologica (JB, GJ) received funding from the International Federation of the Red Cross and Red Crescent Societies to carry out this work.

ascertain the exposure that caused their infection. The number of infections prevented through SDB is significant, yet greater impact would be achieved by early hospitalization of the primary case during acute illness.

Author summary

The care of an individual infected with Ebola virus disease (EVD), their death, funeral, and burial in the community rather than in an Ebola Treatment Center (ETC) poses a serious risk for continued disease transmission. Consequently, SDB is an essential component of EVD outbreak response; however, its impact on transmission is not well understood. During the 2013-2016 EVD epidemic the Red Cross carried out over 50% of the official burials in Guinea, Liberia and Sierra Leone. We performed epidemiological investigations in EVD affected communities to better understand disease transmission linked to unsafe burials of (suspect) EVD infected individuals, and risk factors for transmission linked to caring and burial practices. An average of 2.58 secondary cases were potentially generated per unsafe burial investigated and varied by district (range: 0-20). Additionally, the Red Cross SDB program potentially averted between 1,411 and 10,452 secondary EVD cases, reducing the epidemic by 4.9% to 36.5%. Our results quantify for the first time the potential impact this essential EVD response component had on the 2013-2016 epidemic and highlight the importance of SDB as a fundamental control measure, while also underlining the well-known importance of isolating EVD infected individuals as soon as they show symptoms in order to limit transmission.

Introduction

The 2013–2016 West Africa Ebola virus disease epidemic reached a scale never seen before. Over 28,600 people were infected with Ebola virus disease (EVD) and over 11,000 died [1,2]. Any activity involving direct, unprotected contact between living individuals and an EVD infected individual, living or deceased, are potential mechanisms for onward transmission of the virus. Transmission of EVD occurs when an uninfected individual has direct contact with the blood or body fluids of an EVD infected, symptomatic individual or objects contaminated with their blood or body fluids. Risk of EVD transmission is particularly high during caregiving practices and when bodies are being prepared for funeral rites [3,4]. Consequently, relatives, community members and healthcare workers without appropriate personal protective equipment have an increased risk of infection.

Despite common modes of transmission, the dynamics of the 2013–2016 EVD epidemic varied widely between and within the most affected countries of Guinea, Liberia and Sierra Leone [5,6]. Nevertheless key pillars of the response were systematically implemented across all affected countries. These pillars, common to EVD outbreaks, focused on preventing disease transmission through case management in an Ebola Treatment Center (ETC), contact tracing, social mobilization and safe burials of EVD infected bodies [7,8].

Deaths of EVD positive individuals that occur in the community, outside an ETC, pose a serious risk for continued EVD transmission. The infection status of the deceased is frequently unknown at the time of death and there are many opportunities for contact with the deceased. These can result, for example, from local customs including washing, redressing, and burial of the body [9,10]. When an EVD positive individual dies in the community, both familial

abstention from contact with the deceased, and a safe and dignified burial (SDB) carried out by a trained team, are necessary to prevent burial- and funeral-related transmission. When an EVD infected body is buried safely, onward transmission of the virus as a consequence of burial or funeral rituals should cease. A SDB, in the strictest terms, should result in no direct contact with the infected body after the time of death. Consequently, no secondary cases should result from a safe burial, and onward transmission should be limited.

The importance of SDB as an integral part of reducing EVD transmission and stopping an outbreak is accepted, but not well understood. While work has been conducted to ascertain the impact of individual EVD interventions on disease transmission, and the importance of community deaths in secondary case generation, most of the data come from limited, focal case time series or relies on data on cases and deaths as published by the World Health Organization (WHO) [11–14]. Until this study, neither research focusing specifically on safe burials nor quantification of the direct impact of such activities had been conducted. Using data collected during epidemiological investigations, we estimate the number of secondary cases that were potentially averted by safe burials, and describe risk factors for EVD transmission during funerals and burial rituals (unsafe burials). The potential impact of the SDB program on the 2013–2016 EVD epidemic as a result of activities carried out by the National Societies of the Red Cross in Sierra Leone, Guinea and Liberia, supported by the International Federation of the Red Cross and Red Crescent Societies (IFRC), is also estimated.

Methods

Ethics statement

Authorization to conduct this research was obtained from the Sierra Leone Ministry of Health and Sanitation, the Guinea Ministry of Health and Public Hygiene and the Liberia Ministry of Health and Social Welfare. Informed consent forms were read and explained to each research participant. Due to restrictive infection prevention procedures, consent to participate was given orally in order to avoid contact between the research team and the key informants. Consent to participate was noted by the research team on the consent form. Participation was voluntary and it was made clear that consent could be withdrawn by the participant at any time and for any reason without repercussion.

All epidemiological data were anonymized before being entered into data spreadsheets. All research documents were stored in a locked cabinet or electronically on a password-protected computer with access available only to the research team. Data back-ups were made on external support.

Study setting and population

The Red Cross conducted approximately 50% of all official SDBs in Sierra Leone, 100% in Guinea and 100% in Montserrado County, Liberia during the 2013–2016 epidemic. Data collection was carried out in rural and urban districts in Sierra Leone, Guinea and Liberia where the National Society in each country, supported by the IFRC (hereafter referred to as the Red Cross) were responsible for SDB, as depicted in Table 1 [15–17].

Implementation

Data collection was carried out from June 18, 2015 to August 4, 2015. Interviews were conducted for four days in each district, except for Montserrado County where they were conducted for eight days. To the furthest extent possible, attempts were made to inform the leaders of the selected villages or urban neighborhoods (hereafter referred to as community)of

Country	District*	Population ⁺	Area	First EVD Case
Sierra Leone	Western Area Rural	197,098	544 km ²	August 2014
Sierra Leone	Kambia	313,765	3,108 km ²	September 2014
Sierra Leone	Kailahun	409,520	3,859 km ²	April 2014
Guinea	Guéckédou	405,000	4,400 km ²	December 2013
Guinea	Forécariah	136,000	4,200 km ²	September 2014
Liberia	Montserrado	1,200,000	1,909 km ²	June 2014

Table 1. Population and area estimates for the districts where unsafe burials were investigated.

*The generic term "district" is used to refer to the different administrative divisions in the three countries. The appropriate nomenclature for each country is: Guinea (prefecture), Liberia (county), Sierra Leone (district).

⁺Population calculations are based on 2004 data with an annual 6% growth rate.

https://doi.org/10.1371/journal.pntd.0005491.t001

the arrival of the research team at least one day prior to their visit in order to ensure the availability of key informants and receive the necessary approvals from the village chief and community leaders.

Prior to data collection in each district, a member of the local Red Cross who was familiar with the area and able to translate between English and the local language(s) was trained on the research methodology and data collection tool. Together with the study's epidemiologist, they visited each community selected and met with community leaders and key informants. Key informants were identified by the Red Cross volunteer in collaboration with community members. One primary key informant was purposively selected in each community to provide information on the burial of interest. This person was generally a family member of the deceased or if a relative was no longer present, another individual identified by the community.

Safety

Data collection occurred during periods of active EVD transmission in some districts. All activities were therefore conducted in accordance with strict infection prevention procedures that focused on avoiding contact between the data collection team, key informants and other community members. All members of the data collection team were provided with personal protection equipment while in the field including gum (rubber) boots, chlorine spray bottles and hand sanitizer containing at least 70% alcohol. In addition, team members were given daily reminders on safety procedures to adopt while in the field to ensure their safety.

Sampling frame

The sampling frame was comprised of communities (rural villages and urban neighborhoods) in priority districts identified by the Red Cross that had experienced at least one of the following events: a) cases of EVD; b) community deaths during the period of the epidemic; or c) having a community member hospitalized in an ETC. If this information was unavailable, a list of all communities in the selected district was used in its place.

During selection, one group of 10 communities (G1) was randomly selected from the sampling frame. After communities were selected, key informants were identified in each community and used to ascertain: 1) if any cases of EVD had occurred in the community, and 2) if an unsafe burial of an EVD case had taken place. If the answer to both questions was yes, the community was included in the final list of communities to be visited by the study team. If at least five communities with unsafe burials were not found in G1, another group of 10 communities (G2) was selected from the sampling frame until at least five communities that reported an unsafe burial of an EVD infected individual were identified.

If the history of an unsafe burial was unable to be reconstructed due to loss of key informants in the community or another cause, another unsafe burial was chosen from the list. If more than one unsafe burial related to the same source case took place in the same community, data were collected from one of the burials, generally the first one that took place.

Definitions

An unsafe burial was considered to be a burial of an individual with suspect EVD infection, buried by their community, family, or manipulated after death but prior to the arrival of the SDB team. The first identified suspect EVD case that was buried unsafely was considered to be the primary case for the purposes of further investigation. A contact was defined as any individual who had physical contact with the body of the deceased, their body fluids or their (potentially infected) belongings after death as reported by the key informant. A potential secondary case was defined as a contact that met either the baseline or ceiling definition (see below). These are called "potential" secondary cases because transmission was not directly observed, and it is therefore possible that some secondary cases acquired the infection from a source other than the primary case.

Data collection

Data were collected on a standardized questionnaire, from key informants, regarding the deceased (primary case) and their contacts, by the epidemiologist in collaboration with the local Red Cross member(s). For each primary case, details of age, sex, source of infection, EVD swab confirmation and secondary cases in individuals listed as having had contact with the deceased (contacts) were collected. For each individual reported to have had contact with the deceased, age, sex, type of contact(s) with the body of the deceased, current status (alive/dead) and EVD swab confirmation was collected. In addition, details about family ties and relationship to the deceased were collected in Liberia and Guinea.

Data management

Completed questionnaires were reviewed daily for completeness and data entered into a Microsoft Excel spreadsheet. Data analysis was conducted in Stata 12 (Stata Corp, College Station, TX) and R (R Foundation for Statistical Computing, Vienna, Austria).

Data analysis

The average number of secondary infections potentially caused by an unsafe burial, and risks of secondary infection associated with different types of contact were estimated using two datasets, both extracted from the epidemiological data collected in the field. The first dataset from which results for the baseline estimate were produced consisted of only EVD positive cases (as reported by the key informant), and contacts were limited to those that only had contact with the primary case after death (i.e. caregivers were excluded). The second dataset from which results for the ceiling estimate were produced consisted of all data collected and included any-one who became sick or died following contact with the primary case (before or after death) as reported by the key informant. The number of secondary cases potentially averted was calculated as the average number of potential secondary cases per unsafe burial multiplied by the number of EVD positive burials carried out by the Red Cross. The number of SDB carried out by Red Cross SDB teams, by country, is presented as reported by the Red Cross. The number of SDB that were laboratory positive corresponds to the number of deaths reported from October 2014 to April 2015 in Sierra Leone for which laboratory results were available and matched based on name, age and sex using a computer algorithm (~75% success in matching) in addition to all positive community deaths in Guinea and Montserrado County, Liberia.

Pooled and district specific estimates are presented with medians and interquartile ranges. Estimates of risk factors among contacts of a primary case are presented as odds ratios based on two-sided Fisher's exact test and presented with 95% confidence intervals. In order to construct the "cared for" odds ratio, the baseline case definition was altered to include caregivers otherwise the case definitions for the baseline and ceiling estimates remained the same for the risk analysis, with the baseline estimate excluding caregivers.

Results

Forty-five unsafe community burials were found and investigated (Table 2). In three of the 42 communities (one in Forécariah, Guinea, and two in Montserrado, Liberia), two unsafe burials were investigated. Two unsafe burials were reported to have not resulted in secondary cases (both in Sierra Leone, one in Kambia and Kailahun). At least five unsafe burials were documented in each of the six districts except for Forécariah (Guinea) where only three unsafe burials were investigated.

Across the study, 310 individuals were identified and reported to have had contact with the body of the primary case after death, with an average of seven contacts per unsafe burial. The number of contacts reported from each unsafe burial ranged widely in and across the three countries: 2–19 in Sierra Leone 3–20 in Guinea and 1–12 in Liberia. Contacts were predominately male (60%) with a median age of 40 years (IQR: 30–40).

Secondary cases associated with unsafe burials

Across the study, 203 contacts, 65% of those recorded, were reported to have become sick after having contact with the body of the primary case during an unsafe burial (Table 3). This ranged from 89% in Western Area Rural (Sierra Leone) to 33% in Kambia (Sierra Leone).

	Sierra Leone			Guinea		Liberia	Overall
	Western Area Rural	Kambia	Kailahun	Guéckédou	Forécariah	Montserrado	Total
Communities included, N	6	5	9	8	2	12	42
Burials investigated, N	6	5	9	8	3	14	45
Contacts identified, N	46	18	77	69	24	76	310
Contacts per burial, median [range]	8 [2–13]	2 [2–8]	8 [4–19]	11 [3–20]	5 [4–15]	5 [1–12]	7 [1–20]
Age⁺, median [IQR]	54 [20–48]	35 [30– 50]	45 [35– 55]	40 [33–55]	30 [23–46]	39 [29–55]	40 [30– 40]
Sex, % male	50	72	56	75	58	54	60
Health status known, n (% of N contacts identified)	42 (91.3)	15 (83.4)	77 (100)	69 (100)	75 (98.7)	23 (95.9)	301 (97.1)
Sick, n (% of N contacts identified)	41 (89)	6 (33)	29 (38)	50 (72)	14 (58)	63 (83)	203 (65)
Confirmed EVD infection, n (% of N contacts identified)	33 (71)	5 (28)	7 (9)	4 (6)	5 (21)	24 (32)	78 (25)

Table 2. Description of reported primary case contacts as described by key informants by country and district.

*All percentages rounded to the nearest whole number; categorical variables will not always sum to 100%

⁺ Age was not reported for 57 individuals in Sierra Leone (Western Area Rural = 22, Kambia = 5 and Kailahun = 30)

https://doi.org/10.1371/journal.pntd.0005491.t002

Contact	Reported to have fallen sick after contact, (%)			
	No	Yes		
Before & after death	22 (22)	120 (59)		
After death only	76 (88)	83 (41)		
Total:	98	203		

Table 3. Reported illness after contact with the primary case and timing of contact as reported by key	1
informants.	

https://doi.org/10.1371/journal.pntd.0005491.t003

Forty-one percent (83/203) of individuals that were reported to have fallen sick had contact with the primary case only after death; however, the key informant was not always aware if EVD infection had been confirmed by laboratory testing. The post-contact health status (sick or EVD confirmed) was unknown by key informants for nine contacts. For those whose health status was known, 25% (78/301) were reported to have fallen ill and have had a laboratory confirmed EVD infection (Table 2).

Contact during unsafe burials

Investigations at the household level revealed that contact was not only limited to contact with the primary case after death. Contacts that occurred after death were classified into seven categories (Table 4). Key informants frequently reported that some individuals had more than one type of contact with the primary case.

Individuals who had contact with the primary case after death were frequently reported (46%, 142/310) to have had contact with the same individual during the acute phase of their illness. Twenty-three contacts (7%, 23/310) were reported to have protected themselves when having contact with the primary case during acute illness or after death. Use of protection was most frequently reported in Kailahun (Sierra Leone) (16% of contacts, 12/77) and Montserrado County (Liberia) (11% of contacts, 8/76).

Risk related to contact

The exposure that was most strongly predictive of secondary transmission was having contact with fluids (e.g. blood, vomit) of a primary case, followed by direct physical contact with a primary case during their acute illness (Fig 1).

Table 4. Ty	ces of contact with	primary cases	as reported by key	y informants by	country a	and district.
-------------	---------------------	---------------	--------------------	-----------------	-----------	---------------

	Sierra Leone		Guinea		Liberia	Overall	
	Western Area Rural	Kambia	Kailahun	Guéckédou	Forécariah	Montserrado	Total
Contacts identified, N	46	18	77	69	24	76	310
Care during illness, n (%)	17 (37)	5 (28)	22 (29)	39 (57)	12 (50)	47 (62)	142 (46)
Contact after death: n, (% N)							
Blood/body fluids	0	0	4 (5)	1 (5)	6 (25)	10 (13)	21 (7)
Washed clothes/bedding	6 (13)	2 (11)	5 (6)	12 (17)	2 (8)	13 (17)	40 (13)
Washed body	18 (39)	5 (28)	40 (52)	21 (81)	7 (29)	21 (28)	112 (36)
Transported body	9 (20)	0	16 (21)	0	14 (58)	36 (47)	75 (24)
Burial/funeral rituals	2 (4)	0	14 (18)	18 (26)	11 (46)	41 (54)	86 (28)
Burial of body	19 (41)	12 (67)	25 (32)	11 (14)	11 (46)	32 (42)	110 (35)
Other	6 (13)	0	1 (1)	14 (54)	0	1 (1)	22 (7)
Used protection, n (%)	0	1 (6)	12 (16)	0	2 (8)	8 (11)	23 (7)

https://doi.org/10.1371/journal.pntd.0005491.t004





https://doi.org/10.1371/journal.pntd.0005491.g001

Overall, in the baseline estimate, 9.5% (+/- 1.6% s.e.; n = 45) of contacts became secondary cases, meaning that these contacts had exposures and outcomes that met the baseline case definition (not caregivers, and were reported to have become infected with EVD). In the ceiling estimate, 68% (+/- 2.6% s.e.; n = 45) of contacts became secondary cases, meaning that these contacts either became sick or died after their exposure.

In the baseline estimate, all laboratory confirmed EVD cases reported some form of contact after death. This precludes a finite estimate of the odds ratio associated with contact after death in the baseline estimate (the lower bound of the 95% confidence interval on the estimate was 8.0). In the ceiling estimate, the odds ratio was 0.16 (95% CI: 0.089–0.30). This reflects that many contacts in this estimate were considered EVD positive, but had contact both before and after death.

People who had contact with the primary case before death (caregivers) were, on average, between 2.63 (95% CI: 1.15–4.63, baseline estimate) and 5.97 (95% CI: 3.31–11.17, ceiling estimate) times more likely to become EVD infected, relative to people who only had contact with the primary case after death. For data on fluid contact, a small sample size caused the upper confidence bound to exceed 300, but this is a sampling artifact and should not be interpreted as containing information on risk patterns.

Variation in estimates by place and time

In the baseline estimate, a single unsafe burial was associated with an average of 0.64 (1.46 s.d., n = 45) secondary cases (Fig 2). This varied among districts from 0.125 in Guéckédou (Guinea) to 3.00 in Western Area Rural (Sierra Leone) (Fig 3). In the ceiling estimate, a single unsafe burial was associated with an average of 4.74 (4.27 s.d., n = 45) secondary cases with variation among districts ranging from 1.2 (1.09 s.d., n = 6) in Kambia (Sierra Leone), and 6.25 (7.39 s.



Type of estimate

Fig 2. Estimates and uncertainty in the number of secondary cases caused by an unsafe burial, pooled across districts, for baseline and ceiling estimates. Thick horizontal lines show median estimate, boxes enclose the interquartile range in the estimate (between the 25th and 75th percentile) across primary cases and districts. Capped vertical lines encompass the total range of variability in the estimate.

https://doi.org/10.1371/journal.pntd.0005491.g002

d., n = 6) in Guéckédou (Guinea) (Fig 3). In both the baseline and ceiling estimates there was no statistically significant evidence of systematic variation among districts; just evidence of an abundance of variability, likely reflecting latent heterogeneities in transmission, and the stochastic nature of the underlying disease transmission process.

Red Cross safe and dignified burial program impact

A total of 47,505 SDB were reported to have been carried out by the National Societies in Sierra Leone, Guinea and Liberia from 2013–2016. The number of burials completed varied greatly by country (Table 5). When laboratory results were received and matched with the individuals buried, 2,205 (4.6%) of the bodies buried tested positive for EVD in the laboratory.

The relatively small number of bodies buried in Liberia, compared to Sierra Leone and Guinea, is related to the areas the Red Cross was responsible for SDB. In Sierra Leone and Guinea, the National Societies were responsible for burials in the entire country, while in Liberia the area of intervention was limited to Montserrado County only.

EVD cases averted

Fig 4 presents estimates of the total number of reported burials carried out by the Red Cross in addition to estimates of the number of these burials that were EVD positive, contrasted with the reported cases from the WHO patient database for Sierra Leone, Liberia and Guinea.



Fig 3. Estimates and uncertainty in the number of secondary cases caused by an unsafe burial, by district. Boxes enclose the interquartile range of each estimate (extending from the 25th percentile to the 75th percentile). The thick horizontal lines within each box correspond to the median. Capped vertical lines encompass the total range of variability in the estimate.

https://doi.org/10.1371/journal.pntd.0005491.g003

Applying the analysis presented above to the number of positive EVD cases buried, the number of secondary cases potentially caused by a typical unsafe burial leads to estimates for the number of secondary cases potentially averted by the SDB program. Using this method we estimate that between 1,411 (lower estimate) and 10,452 (upper estimate) secondary EVD cases may have been prevented by the SDB program. This represents a reduction in the total size of the 2013–2016 EVD epidemic of between 4.9% (lower estimate) and 36.5% (upper estimate). Thus, the total number of EVD cases in the 2013–2016 epidemic could have been 30,012 to 39,053 if the SDB program had not been implemented by the Red Cross in Sierra Leone, Guinea and Liberia.

Discussion

The majority of the unsafe burials investigated seemed to be relatively representative of classic EVD transmission with between 0.64 to 4.74 secondary EVD cases associated with unsafe burials, an overall average of 2.58. Estimates of the average number of secondary cases potentially resulting from an unsafe burial are significant although there is variation depending on which estimate, baseline or ceiling, was used. This variation and the use of two estimates highlight the uncertainty surrounding the dynamics of EVD transmission. Additionally, both the number

Table 5. Number of safe and dignified burials carried out by the Red Cross by country and the number of those deaths that were EVD positive.

Country	Reported SDB completed	EVD positive		
Sierra Leone	26,308	1,413		
Guinea	17,513	254		
Liberia	3,684	538		
Total:	47,505	2,205		

https://doi.org/10.1371/journal.pntd.0005491.t005



Fig 4. Reported cases in the WHO patient database and burials over time in Sierra Leone (left), Liberia (center) and Guinea (right). Colored dots show reported cases from the WHO patient database. Dotted lines show the total number of recorded burials carried out. Solid lines show estimates of the number of burials of EVD casualties, reconstructed from laboratory testing of swab samples collected from community deaths.

https://doi.org/10.1371/journal.pntd.0005491.g004

of individuals reported to have had contact with the primary case, and the type of contact they had during an unsafe burial, ranged widely by district. This variation may be due to many factors including the incidence of EVD in the district and the point in the epidemic during which the unsafe burial occurred. Accordingly, the highest median number of contacts per burial in this research was reported in Guéckédou (Guinea), the first district affected by the EVD epidemic, and the lowest in Kambia (Sierra Leone), the last of the sampled districts to be affected. This variation may reflect changes in behavior after exposure to EVD related messaging by Red Cross and other partners, and familiarity of individuals and communities with the disease.

The type of contact that was had with a primary case and the timing of that contact, before or after death, are important factors in EVD transmission. Such information is normally collected during field outbreak investigations for the purposes of contact tracing and to ascertain epidemiological links to known cases. In the early days of the EVD epidemic, analysis of contact tracing data showed that of 701 confirmed EVD cases 67 (10%) reported contact with EVD infected individuals after death (only), while 148 (21%) reported contact both before and after death [18]. Furthermore, some of the potential secondary cases identified in our study may have been infected by a source other than the primary case. In these cases, preventing exposure to the primary case would still reduce the risk of onward transmission, but would not be guaranteed to avert the secondary case altogether. Risk estimates from data presented here show, similar to previous studies, that having contact with the primary case both during illness and after death significantly increased the risk of EVD transmission compared to individuals who had contact with the primary case only after death [3,14,19]. Further analyses of risk factors for EVD transmission highlight the importance of having direct contact with the blood or body fluids of the deceased. Although our sample size was too small to provide a reliable estimate of risk due to such contact, exposure to blood and body fluids of the deceased was frequently cited by key informants and cannot be discounted as an important factor in transmission, as documented in previous risk factor analyses [3,19].

Reports of protective measures taken while having contact with a primary case varied in frequency and method, reflecting distinct differences in EVD awareness and behavior. Knowledge, attitudes and practices (KAP) surveys carried out at different points in the epidemic also highlighted geographic differences regarding EVD and its transmission [20,21]. Significant differences were noted in areas (defined differently between surveys) that had a

higher incidence of EVD cases when compared to areas with lower EVD case incidence. While data collected through KAP surveys are subject to limitations, the differences between areas remains illustrative.

Not only are the processes that control changes in the number of individuals infected with EVD by place or over time not well understood, the number of individuals infected during the West African epidemic is also not certain due to significant underreporting [2,22]. Furthermore, neither burials carried out by other organizations during the epidemic, nor burials conducted early in the epidemic by the Red Cross are taken into consideration in this estimation. Systematic data collection by Red Cross burial teams only began later in the response (July 2014 in Liberia, October 2014 in Sierra Leone, and January 2015 in Guinea), laboratory data were difficult to obtain in some areas and data from other organizations involved in SDB were unavailable resulting in an underestimation of both overall SDB program impact and the impact of the Red Cross SDB program in particular.

The impact of an SDB program is highly dependent on disease prevalence at the time of program initiation. As seen in the data from Guinea, it appears that initiating SDB later in the epidemic when there are fewer EVD cases results in a dramatic reduction in program impact because the bodies buried will primarily be EVD negative. The Red Cross burial program began early in April 2014 in Guinea; yet, systematic collection of samples from the deceased only began in January 2015 when the prevalence of EVD was lower. As a result, the Red Cross program in Guinea is estimated to have buried mostly EVD negative bodies as reflected in the low numbers of confirmed laboratory positive bodies. This contrasts with Liberia where less than 20% of t burials were carried out by the Red Cross; however, twice the number of positive bodies were buried. This is likely a consequence of the high prevalence of EVD when SDB was initiated in Liberia, and an EVD positivity that was higher overall.

The impact of SDB on preventing chains of transmission was not investigated and may have been particularly relevant in the first months of the epidemic when the reproductive number was greater than one [5,18] and the epidemic was spreading quickly in communities. Furthermore, the level of knowledge about EVD varied over the course of the epidemic, as did communities' sense of fear and/or distrust of EVD responders [9,10,20,21,23–27]. In such situations, each secondary case averted could be the first of many cases in an averted chain of transmission likely composed of multiple transmission generations and secondary cases. In Liberia, for example, one to seven transmission generations were documented in a single transmission chain with 4–35 secondary cases produced [14]. The estimates presented here correspond to the first transmission generation after the primary case and do not consider the impact that SDB had on chains of transmission averted by preventing cases beyond the first transmission generation. Consequently, the number of cases averted by SDB programs may be considerably higher than that which was directly estimated.

None of the investigations of unsafe burials of suspect EVD infected individuals reported here document a superspreading event, an event such as an unsafe burial that results in a disproportionately high number of secondary transmission (infection) to contacts. Nevertheless, SDB may have had an even greater impact on the epidemic by preventing superspreading events. Indeed, modeling suggests that such events may have sustained EVD transmission in Sierra Leone [28]. These events have been infrequently documented but have been known to occur after the death of an influential person in the community (e.g. elder or traditional healer), such as in Kissidougou (Guinea) where an unsafe burial of a male midwife assistant responsible for circumcisions in the community, was reported to have caused upwards of 85 secondary cases [29,30]. While superspreading events are relatively rare, if SDB prevented even one superspreading event, the impact on averted chains of transmission would be significant. Although such events were not identified during this research, both anecdotal reports and reports in peer reviewed literature suggest that superspreading events associated with unsafe burials did occur during the 2013–2016 outbreak [29,31].

The data presented here do have a number of limitations, including recall bias. As unsafe burials were investigated, the primary case was deceased and all information was collected from key informants. Age of the contact was frequently unknown and unable to be estimated with any certainty, particularly by key informants external to the family unit. Data were collected retrospectively and occasionally the unsafe burial investigated had occurred more than 12 months prior to the date of the interview, as was the case in Guéckédou (Guinea) and Kailahun (Sierra Leone). Conversely, as in Kambia and Western Area Rural (Sierra Leone), some unsafe burials had recently occurred and interviews with key informants were conducted while the individuals were still in quarantine. Frequently, key informants had themselves been infected with EVD and hospitalized due to the unsafe burial that was being investigated. In these cases it was difficult to collect data on the health outcomes of individuals who had participated in the unsafe burials if the outcome occurred during the hospitalization of the key informant. In some instances information could not be collected from familial key informants as they had died prior to the visit of the study teams. In such cases, additional key informants were identified. Furthermore, the risk factor analysis is limited by the amount of data that we were able to collect from key informants. Recall of caregiving and funeral practices was sometimes limited or not available. In addition, the status of some contacts, sick or EVD confirmed, was unknown if they did not return to their communities or share the information. In the absence of laboratory confirmation of EVD infection for all contacts reported to have become cases (fallen sick or had an EVD infection), we only documented the first transmission generation from the identified primary case. To the furthest extent possible, however, we attempted to triangulate the information collected during interviews in order to ensure the quality of the data and ensure it was exhaustive as possible. Finally, not all laboratory results were available or could be matched to the burials carried out, and the period for which results were available was limited.

Here, post-mortem transmission of EVD is investigated exclusively and the implications evaluated systematically for the first time. These findings underscore the substantial impact that SDB had on the 2013–2016 EVD epidemic with regards to the number of cases averted and the importance of early initiation of SDB activities during an EVD response. Previously, the importance of SDB as an integral part of reducing EVD transmission and stopping epidemics was acknowledged as a fundamental EVD control measure in terms of its impact on ending transmission, however its impact was not well understood.

The timing of SDB implementation and scale-up is also important and will affect the magnitude of the outbreak. The earlier in an EVD outbreak that a SDB program can be implemented at scale, the greater the number of cases averted and consequently the larger the program's impact on the outbreak will be. Additional research is needed in order to better understand the variation of risk for EVD transmission related to distinct care practices both before and after death. The SDB program is a cornerstone of comprehensive EVD prevention and response programs and must be implemented as a holistic intervention incorporating both public health and socio-cultural perspectives in order to enable quick scale up and achieve maximum impact.

Supporting information

S1 Checklist. This file includes a checklist of information to include when reporting data from observational studies, indicating where specific items are reported in the paper. (DOC)

Acknowledgments

We acknowledge the support of the Red Cross National Societies of Guinea, Sierra Leone and Liberia which greatly facilitated and contributed to this research.

Results from this research were presented at the annual meeting of the American Society for Tropical Medicine and Hygiene in Atlanta, Georgia, 2016.

Author Contributions

Conceptualization: AT AM BDD GJ JB.

Data curation: AT BDD HKN.

Formal analysis: BDD AT.

Funding acquisition: AM.

Investigation: AT GJ HKN RNB DJ AW.

Methodology: AT AM BDD GJ JB.

Supervision: AT GJ HKN RNB DJ AW.

Visualization: AT BDD.

Writing - original draft: AT.

Writing - review & editing: AM BDD GJ JB HKN RNB DJ AW.

References

- 1. World Health Organization. Ebola Situation Report—6 January 2016 [Internet]. 2016 [cited 2015 Nov 12]. http://apps.who.int/ebola/current-situation/ebola-situation-report-6-january-2016
- WHO Ebola Response Team. After Ebola in West Africa—Unpredictable Risks, Preventable Epidemics. N Engl J Med. 2016; 375(6):587–96. https://doi.org/10.1056/NEJMsr1513109 PMID: 27509108
- Dowell SF, Mukunu R, Ksiazek TG, Khan AS, Rollin PE, Peters CJ. Transmission of Ebola hemorrhagic fever: a study of risk factors in family members, Kikwit, Democratic Republic of the Congo, 1995. Commission de Lutte contre les Epidémies à Kikwit. J Infect Dis. 1999; 179(Supplement 1):S87–91.
- Osterholm MT, Moore KA, Kelley NS, Brosseau LM, Wong G, Murphy FA, et al. Transmission of Ebola Viruses: What We Know and What We Do Not Know. MBio. 2015; 6(2):e00137–15. <u>https://doi.org/10. 1128/mBio.00137-15</u> PMID: 25698835
- 5. Althaus CL. Estimating the Reproduction Number of Ebola Virus (EBOV) During the 2014 Outbreak in West Africa. PLoS Curr Outbreaks. 2014 Sep 2(Edition 1).
- Fang L, Yang Y, Jiang J, Yao H, Kargbo D, Li X, et al. Transmission dynamics of Ebola virus disease and intervention effectiveness in Sierra Leone. Proc Natl Acad Sci. 2016; 113(16):4488–93. <u>https://doi.org/10.1073/pnas.1518587113</u> PMID: 27035948
- 7. World Health Organization. Ebola and Marburg virus disease epidemics: preparedness, alert, control, and evaluation. Geneva; 2014.
- 8. World Health Organization. Ebola response: package and approaches in areas of intense transmission of Ebola virus. Geneva; 2014.
- 9. Fairhead J. The Significance of Death, Funerals and the After-Life in Ebola-Hit Sierra Leone, Guinea and Liberia: Anthropological Insights into Infection and Social Resistance (Draft). Ebola Response Anthropology Platform. [Internet]. Ebola Response Anthropology Platform. 2014 [cited 2015 May 15]. http://www.ebola-anthropology.net/key_messages/the-significance-of-death-funerals-and-the-after-life-in-ebola-hit-sierra-leone-guinea-and-liberia-anthropological-insights-into-infection-and-social-resistance/
- Fairhead J. Understanding Social Resistance to Ebola Response in Guinea. [Internet]. Ebola Response Anthropology Platform. 2015 [cited 2015 May 10]. http://www.ebola-anthropology.net/wp-content/ uploads/2015/04/Fairhead-EbolaASRFinalSubmissionWeb.pdf
- Ajelli M, Parlamento S, Bome D, Kebbi A, Atzori A, Frasson C, et al. The 2014 Ebola virus disease outbreak in Pujehun, Sierra Leone: epidemiology and impact of interventions. BMC Med. 2015; 281(13).

- 12. Rivers CM, Eubank S. Modeling the Impact of Interventions on an Epidemic of Ebola in Sierra Leone and Liberia. PLoS Curr Outbreaks. 2014 Nov 6(Edition 2).
- Shen M, Xiao Y, Rong L. Modeling the effect of comprehensive interventions on Ebola virus transmission. Sci Rep. 2015; 5:15818. https://doi.org/10.1038/srep15818 PMID: 26515898
- Lindblade KA, Kateh F, Nagbe TK, Neatherlin JC, Pillai SK, Attfield KR, et al. Decreased Ebola Transmission after Rapid Response to Outbreaks in Remote Areas, Liberia, 2014. Morb Mortal Wkly. 2015; 21(10).
- 15. Statistics Sierra Leone. Sierra Leone [Internet]. 2015 [cited 2015 Nov 1]. http://www.geohive.com/cntry/ sierraleone.aspx
- Liberia Institute of Statistics and Geo-Information Service. Liberia [Internet]. 2015 [cited 2015 Nov 1]. http://www.geohive.com/cntry/liberia.aspx
- Institut National de la Statistique Guinée. Guinea [Internet]. 2015 [cited 2015 Nov 1]. p. 2014. <u>https://www.cia.gov/library/publications/the-world-factbook/geos/gv.html</u>
- WHO Ebola Response Team. Ebola Virus Disease in West Africa—The First 9 Months of the Epidemic and Forward Projections. N Engl J Med. 2014; 371(16):1481–95. <u>https://doi.org/10.1056/ NEJMoa1411100 PMID: 25244186</u>
- Francesconi P, Yoti Z, Declich S, Onek P a, Fabiani M, Olango J, et al. Ebola hemorrhagic fever transmission and risk factors of contacts, Uganda. Emerg Infect Dis. 2003; 9(11):1430–7. https://doi.org/10.3201/eid0911.030339 PMID: 14718087
- 20. United Nations Children's Educational Fund (UNICEF), Catholic Relief Services, Focus 1000. Study on public knowledge, attitudes and practices relating to Ebola Virus Disease (EVD) Prevention and medical care in Sierra Leone. Freetown; 2014.
- 21. Kobayashi M, Beer KD, Bjork A, Chatham-Stephens K, Cherry CC, Arzoaquoi S, et al. Community Knowledge, Attitudes and Practices Regarding Ebola Virus Disease—Five Counties, Liberia, September-October, 2014. Morb Mortal Wkly Rep. 2015; 64(26):2002–13.
- World Health Organization. Why the Ebola outbreak has been underestimated [Internet]. World Health Organization Media Centre. 2014. http://www.who.int/mediacentre/news/ebola/22-august-2014/en/
- 23. Niang Cl. Anthropological Study of Resistances to Ebola Response Interventions in Guinea: Preliminary Findings and Recommendations. WHO Mission (February-March 2015). 2015.
- Peters MM. Community Perceptions of Ebola Response Efforts in Liberia: Montserrado and Nimba Counties. Oxfam GB. 2014.
- 25. Wigmore R. Ebola and Rumours in Sierra Leone: Contextualising Ebola Rumours from a Historical, Political and Social Perspective. University of Sussex; 2015.
- Wilkinson A, Leach M. Briefing: Ebola—Myths, Realities, and Structural Violence. Afr Aff (Lond). 2014;1–13.
- 27. Johnson G, Bedford J, McClelland A, Tiffany A, Dalziel BD. Evaluating the impact of Safe and Dignified Burials for stopping Ebola transmission in West Africa: summary findings from the anthropological study in Guinea, Sierra Leone and Liberia. Geneva, Switzerland; 2015 (unpublished).
- Lau MSY, Dalziel BD, Funk S, McClelland A, Tiffany A, Riley S, et al. Spatial and temporal dynamics of superspreading events in the 2014–2015 West Africa Ebola epidemic. Proc Natl Acad Sci [Internet]. 2017;pii: 20161:[Epub ahead of print]. http://www.pnas.org/lookup/doi/10.1073/pnas.1614595114
- Victory KR, Coronado F, Ifono SO, Soropogui T, Dahl BA. Ebola Transmission Linked to a Single Traditional Funeral Ceremony—Kissidougou, Guinea, December, 2014-January 2015. Morb Mortal Wkly. 2015; 64(14).
- Althaus CL. Ebola superspreading. Lancet Infect Dis. 2015; 15(5):507–8. https://doi.org/10.1016/s1473-3099(15)70135-0 PMID: 25932579
- Wauquier N, Bangura J, Moses L, Squire J, Roberts W, Gonzalez JP. Understanding the Emergence of Ebola Virus Disease in Sierra Leone: Stalking the Virus in the Threatening Wake of Emergence. PLoS Curr Outbreaks. 2015 Apr 2(Edition 1).